Physics G4003 – Advanced Mechanics

• Instructor: Lam Hui • Class meeting times and place: MW 9:35 am - 10:50 am, Pupin 420 • Office hours: MW 11 am - noon, Pupin 1013 • Phone: 212-854-6892. Email: lhui@astro.columbia.edu • Teaching Assistant: Andrei Savici. Email: andrew@phys.columbia.edu.

Weekly recitations will be arranged.

• Textbooks. Required: Classical Dynamics by Marion Thornton. Recommended: Mechanics by L. D. Landau and E. M. Lifshitz.

• Course requirements: Problem sets (50%), mid-term (25%) and final exam.

(25%). Mid-term is in class on 3/13/02, and final exam. is on 5/15/02 morning.

• Class information can be found at : http://www.astro.columbia.edu/∼lhui.